Author's response to reviews

Title: Gitelman-like syndrome after cisplatin therapy: a case report and literature review.

Authors:

Kessarin Panichpisal (kessarin.panichpisal@ttuhsc.edu)
Freddy Angulo-Pernett (Freddy.angulopernett@ttuhsc.edu)
Sharmila Selhi (Sharmila.selhi@ttuhsc.edu)
Kenneth M Nugent (kenneth.nugent@ttuhsc.edu)

Version: 4 Date: 15 May 2006

Author's response to reviews: see over
May 12th, 2006

Re: Revised Manuscript # 9034259818886939

Dear BMC nephrology editors

Thank you very much for accepting our manuscript for publication in your journal. We did revised the format of manuscript as you requested

1) In abstract section, we change case report to case presentation.
2) In manuscript section, we include the acknowledgements section: “Written consent was obtained from the patient for publication of study”.

Please contact us if you need further information. Thank you.

Sincerely

Kenneth Nugent MD
Kessarin Panichpisal MD